Author’s response to reviews

Title: Assessing the exertion required to induce breathlessness in a population with advanced cancer: Matching measures to the level of physical function

Authors:

Kahren White (whitekahren@gmail.com)

Meera Agar (meera.agar@uts.edu.au)

David Currow (david.currow@uts.edu.au)

Version: 2 Date: 18 Dec 2018

Author’s response to reviews:

Please refer to Response to Reviewer letter uploaded as supplementary material